Author's response to reviews

Title: Knowledge and experience sharing practices among health care professionals in hospitals under Addis Ababa health bureau, Ethiopia

Authors:
Mulusew Mr Andualem (muler.hi@gmail.com)

Version: 2 Date: 30 May 2014

Author's response to reviews:

This article is the first article in the study area. The findings are so vital in showing the current status of knowledge and experience sharing practices in the study area. Based on this result, most of the respondents need to share their knowledge and experience to their staffs, but it is almost none practical due to several reasons. Health professionals are vital components in delivering quality health care services to their customers. To do this, they should be updated to recent health related information and technologies. Therefore, this finding will serve as a baseline source for, health care professionals, health officials, and other concerned bodies plan and take action on the issue.

The best way to disseminate such valuable resources is publishing recent findings at open resource publishing sites. to do this, BMC publishing site is an ideal place, that is why I need to send this important manuscript to this site to be published and accessed openly.